
BEAST TPC Status
- Nene removed from FWD tunnel (either 

trigger not working or cable messed up)

- Humu now has sources from Nene and is in 
the FWD tunnel, but HV wasn’t stable

- Nene brought back to UH

- FPGA is resetting in the FWD DAQ, BWD is 
stable

- FPGA was also having trouble connecting to 
Humu after replacing thermal paste (but was 
working fine before). This could be cable 
related 1



BEAST TPC Plans
- Nene will get new sources and testing at UH 

(to confirm whether this is trigger or cable 
related)

- Data analysis of existing TPC data can 
continue

- Qingyuan and I will work on getting the DAQ 
going again, and he will visit the IR to make 
sure the HV cables are correctly configured

- Once Humu is functional, BWD and FWD 
TPCs should both be ready for new data once 
the machine is on
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